Author’s response to reviews

Title: Exploring variations in childhood stunting in Nigeria using league table, control chart and spatial analysis

Authors:

Victor T Adekanmbi (vkanmbi@yahoo.com)
Olalekan A Uthman (uthlekan@yahoo.com)
Oludare O Mudasiru (mostdare.depthconsult@hotmail.com)

Version: 5 Date: 11 April 2013

Author’s response to reviews:

11th April 2013

The Editorial Team:

Re: Exploring variations in childhood stunting in Nigeria using league table, control chart and spatial analysis

It is certainly a wonderful news that our manuscript is now accepted for publication. We are pleased to submit a revised manuscript referenced above according to the editorial team’s comment.

In the following page we provide a point-by-point response to the editorial team’s comment. We have put the comment in bold with response below each comment.

We would like to thank you in advance for your consideration.

Yours faithfully,
Dr Victor Adekanmbi
For Authors

Editorial comment:

Thank you for submitting your manuscript with the title (above). I am delighted to inform you that we intend to accept your manuscript for publication in BMC Public Health. However, we would need you to revise your Author’s Contributions section so that each author is specifically mentioned.

Authors’ reply: We have now revised the Author’s Contribution section according to BMC Public Health format as below and we hope you find it acceptable:

“VTA and OAU conceived the study. VTA and OAU completed all statistical analyses. VTA, OAU and OOM drafted the manuscript. VTA, OAU and OOM contributed to the discussion. VTA and OAU revised the manuscript. All authors
have read and approved the final manuscript"